Author's response to reviews

Title: Narrow-band imaging does not improve detection of colorectal polyps when compared to conventional colonoscopy: a randomized controlled trial and meta-analysis of published studies.

Authors:

Luis C Sabbagh (lsabbagh@colsanitas.com)
Ludovic Reveiz (mmreveiz@hotmail.com)
Diego Aponte (didimauro673@yahoo.es)
Sylvia de Aguiar (sylvia_deaguiar@hotmail.com)

Version: 2 Date: 1 March 2010

Author's response to reviews:

The Editor

BMC Gastroenterology

Re: Submission of manuscript for publication

Dear Sir,

We have include the link to the LATINREC registry. We also revised the manuscript so that the results are reported according to the guidelines of PRISMA and CONSORT. However, taking into account that we are submitting a trial and a systematic review in a same paper, we have some space limitations, particularly in the abstract.

As mentionned in an email, we cannot afford any payment. I have made a mistake by sending the manuscript without any complete waiving.

The manuscript has been prepared in accordance with the publisher’s requirements. The authors declare that they have no competing interests. The Corresponding Author has the right to grant on behalf of all authors.

Yours sincerely,

Ludovic Reveiz MD MSc.